Author’s response to reviews

Title: Diabetes related knowledge among residents and nurses: A multicenter study in Karachi, Pakistan.

Authors:

Asma Ahmed (asma.ahmed@aku.edu)
Abdul Jabbar (abdul.jabbar@aku.edu)
Lubna Zuberi (lubahamzub@hotmail.com)
Muhammad Islam (muhammad.islam@aku.edu)
Khusro Shamim (khusro.shamim@aku.edu)

Version: 6 Date: 18 July 2012

Author’s response to reviews:

The manuscript has been modified additinally based on Editor and reviewer comments.

1. The reason for using content analysis strategy was that some of the answers were descriptive. The method has been described in an analysis section. The coding of the questions has been described in method section as advised by reviewer.

2. Since it was a pre-validated questionnaire, the test-retest reliability of the questionnaire has been mentioned. Further more, the content validity has been described the way it was carried out in method section.